Author's response to reviews

Title: A randomized controlled trial of telemonitoring in older adults with multiple chronic conditions: the Tele-ERA study

Authors:

Paul Y Takahashi (takahashi.paul@mayo.edu)
Gregory J Hanson (hanson.gregory@mayo.edu)
Jennifer L Pecina (pecina.jennifer@mayo.edu)
Rajeev Chaudhry (chaudhry.rajeev@mayo.edu)
Nilay D Shah (shah.nilay@mayo.edu)
James M Naessens (naessens@mayo.edu)

Version: 4 Date: 26 August 2010

Author's response to reviews:

To editors of BMC:

Thank you for your editorial comments. We have made changes as noted.

Title: please ensure that the title is identical in the manuscript and the submission system

We changed the title of the manuscript to reflect the title submitted.

Authors' contributions: Please indicate that all authors read and approved the final manuscript.

We have placed the statement in the manuscript as all authors have approved of the manuscript.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Tables- please remove the visible vertical lines from your tables

We did remove the lines from tables one and two.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.